**Author's response to reviews**

**Title:** Pulsed Electromagnetic Fields in the treatment of fresh scaphoid fractures. A multicenter, prospective, double blind, placebo controlled, randomized trial

**Authors:**

- Pascal Hannemann (p.hannemann@mumc.nl)
- Kevin W.A. Göttgens (k.gottgens@me.com)
- Bob J. van Wely (bobvanwely@yahoo.com)
- Karel A. Kolkman (kkolkman@alysis.nl)
- Andries J. Werre (a.werre@cwz.nl)
- Martijn Poeze (m.poeze@mumc.nl)
- Peter R.G. Brink (p.brink@mumc.nl)

**Version:** 3  **Date:** 16 April 2011

**Author's response to reviews:**

Dear sir/madam,

We have made the following changes based on your e-mail

Major revisions

Abstract: Added a 'Discussions' section after 'Methods/Design'.
Manuscript Structure: Included a 'Discussions' section after 'Methods/Design' in the main manuscript text.

Minor revisions

Highlighting: removed any colored text and underlining
Typeography: Correct some minor typographic errors.

We would be glad to provide any more information or answer questions about our manuscript.

Sincerely,

P. Hanneman, K. Göttgens, B. van Wely, K. Kolkman, A. Werre, M. Poeze, P. Brink